Author's response to reviews

Title: Knowledge of obstetric danger signs and birth preparedness practices among women in rural Uganda

Authors:

Jerome K Kabakyenga Dr (jerome.kabakyenga@med.lu.se)
Per-Olof Östergren Prof (per-olof.ostergren@med.lu.se)
Eleanor Turyakira Ms (eleanor.tk@gmail.com)
Karen O Pettersson Dr (karen.odberg_pettersson@med.lu.se)

Version: 3 Date: 15 November 2011

Author's response to reviews: see over
Dear Editor,

Subject: SUBMISSION OF REVISED MANUSCRIPT- VERSION 3

I am enclosing herewith a manuscript entitled “Knowledge of obstetric danger signs and birth preparedness practices among women in rural Uganda – Version 3”.

We have responded to all points raised by the Editorial Production Team (see page 2 & 3).

All authors have read through the revised manuscript and have consented to its submission.

Yours Sincerely

Jerome K. Kabakyenga

Corresponding Author
Formatting changes to be made:
---------------------------------
Major revisions (we require the author to make these changes)

Author list: The list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs).

The list of authors in the manuscript is written exactly as they are in the submission system.

Abstract: This should be identical in the manuscript and the submission system. Please remove the key words list from the submission system.

The Abstract is identical in the manuscript and the submission system

References: Please provide a functioning link to reference 23.

The website which had been given has been hacked into. A five-year strategic plan for Mbarara District Local government report is now given as the reference.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Qualifications: Please remove all qualifications/titles from the author list on the submission system.

The titles have been removed in the submission system

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript. Please ensure all text is black.

All track changes have been removed

Abbreviations: Please format these in to a paragraph. Please move this paragraph immediately after the conclusions.
All abbreviations have been formatted to a paragraph and this paragraph has been placed after conclusions.

Authors’ contributions: Please ensure this is formatted into a paragraph rather than a list or in bullet points.

Authors’ contributions have been formatted into a paragraph

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

The manuscript has been checked for typographical errors.